Author's response to reviews

Title: An open-label study examining the effect of pharmacological treatment on mannitol- and exercise-induced airway hyperresponsiveness in asthmatic children and adolescents with exercise-induced bronchoconstriction

Authors:

Salome Schafroth Toeroek (salome.schafroth@usb.ch)
Thomas Mueller (thm.mueller@gmail.com)
David Miedinger (david.miedinger@unibas.ch)
Anja Jochmann (anja.jochmann@ukbb.ch)
Ladina Joos Zellweger (ladina.joos@claraspital.ch)
Sabine Sauter (sabinesautter@yahoo.de)
Alexandra Goll (Alexandra_Goll@gmx.net)
Prashant N Chhajed (pchhajed@gmail.com)
Anne B Taegtmeyer (anne.taegtmeyer@usb.ch)
Bruno Knoepfli (bruno.knoepfli@k-i-m.ch)
Joerg D Leuppi (joerg.leuppi@ksli.ch)

Version: 2
Date: 25 May 2014

Author's response to reviews:

1. Table 1 has been changed. Sex and individual age was removed from the table.
2. One email address has been changed
3. Line and page numbering has been added to the manuscript